Author's response to reviews

Title: Proteomic approach could help us in diagnosis of Riedel's thyroiditis: a case report

Authors:

Pietro Iacconi (p.iacconi@dc.med.unipi.it)
Laura Giusti (giusti@farm.unipi.it)
Ylenia Da Valle (ylenia.davalle@gmail.com)
Federica Ciregia (ciregia@farm.unipi.it)
Gino Giannaccini (ggino@farm.unipi.it)
Liborio Torregrossa (liboriotor@yahoo.it)
Agnese Proietti (a.proietti@dc.med.unipi.it)
Gianluca Donatini (giacko76@hotmail.com)
Salvatore Mazzeo (s.mazzeo@med.unipi.it)
Fulvio Basolo (f.basolo@med.unipi.it)
Antonio Lucacchini (lucas@farm.unipi.it)

Version: 2 Date: 23 November 2011

Author's response to reviews: see over
To Editor in Chief
of Journal Medical Case Reports

Dear Editor,

according to your suggestions the Title of the manuscript (Journal of Medical Case Reports
manuscript: 1891600987636738) has been changed in “Proteomic approach could help us in
diagnosis of Riedel’s thyroiditis: a case report”. Moreover, I assert that this is the first time ever
proteomic analysis has been used in Riedel's thyroiditis to evaluate the possibility of its use in
assisting in diagnosis of this disease.

Sincerely Yours

Antonio Lucacchini